Author's response to reviews

Title: Development and evaluation of a cancer-related fatigue patient education program: protocol of a randomized controlled trial

Authors:

Nina N Stuhldreher (nstuhldreher@uni-bremen.de)
Karl K Reif (karlreif@uni-bremen.de)
Ulrike U de Vries (udevries@uni-bremen.de)
Stefan S Goerres (sgoerres@uni-bremen.de)
Franz F Petermann (fpeterm@uni-bremen.de)

Version: 2 Date: 14 July 2008

Author's response to reviews:

Information about obtaining written informed consent of eligible patients has been added in the section Methods/Summative Evaluation/Identification of eligible patients.